EV/Week 38 starting 18sept 06
SM18 Operation Meeting (Tuesdays) 19 sept 06

Bench status  on Monday  18-09-2006 at 9/18/2006 10:37:51 AM
	5
	A1
	MBAR1334
	PT 2 HV Cold Test
	1.90

	9
	A2
	MBAR1375
	ICS 4 Final connection
	296.97

	4
	B1
	SSS134
	ICS 6 Disconnect Magnet WP07
	298.04

	6
	B2
	SSS139
	PT 3 QH Measure
	1.91

	7
	C1
	SSS338
	PT 3 QH Measure
	2.31 HV

	1
	C2
	SSS357
	PT 12 Warm Up
	12.85

	12
	D1
	SSS302
	PT 14 Magnet MAPS
	296.88

	10
	D2
	SSS608
	PT 13.2 Resist. Meas.
	296.34

	8
	E1
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	298.58

	3
	E2
	MBAL2352
	PT 13.1 IAP @ Warm
	300.28

	11
	F1
	MBAL2280
	ICS 4 Final connection
	295.98

	2
	F2
	MBBR1342
	PTE Min. NRJ 3KA Diode Test
	4.42 HV


 
  
CONFIG :  D 6 - AS 5 -  IRSSS 1 ;   
Supervisor:  Sandrine  ;Eq support : 16-3602 
17 Magnet Tests completed ( 14 Magnet Stripped ) 
8 Dipoles tests  ( MM for 2274 and 1400)


5 Arc-SSS  tests (ROBTC : 339 (Stand By) and 298 (Stripping))
 1  500-series [ 526 (192 hr) with thermal cycle]
0  600-series
General Points:
· News /Situation  : Configuration change for S5/S6, less production for the moment.
· Situation from Cryogenics:  J-P Lamboy or ACR Rep.
· Situation SSSS  :D1 & D2 issues; 
· A.O.B..
· Printers problem in SM18 Control room (Tim Smith and Pierre Charrue informed)
· CQR S6 and S5 has been change because of wrong value.
· WP04 Problems on several bench (A2 and F1).
· 243 NC for circuit open.
	Magnets under Test from 10-09-2006 to 16-09-2006 

	-
	Bench
	Magnet
	Shafts
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	
	

	1
	TBA1
	MBAR2274(2) 
	[image: image3.png]



	09-09-2006 
	30 h.
	32 h.
	34 h.
	16 h.
	15 h.
	14-09-2006 
	127 h.
	Stripping
	
	
	

	2
	TBA2
	MBAR1338 
	[image: image4.png]



	08-09-2006 
	31 h.
	22 h.
	22 h.
	26 h.
	11 h.
	12-09-2006 
	112 h.
	Stand By
	
	
	

	3
	TBB1
	SSS329 
	[image: image5.png]



	05-09-2006 
	24 h.
	13 h.
	107 h.
	72 h.
	24 h.
	15-09-2006 
	240 h.
	Stripping
	
	
	

	4
	TBB2
	SSS243 
	[image: image6.png]



	11-09-2006 
	33 h.
	16 h.
	35 h.
	8 h.
	26 h.
	16-09-2006 
	118 h.
	Stand By
	
	
	

	5
	TBB2
	SSS309(2) 
	[image: image7.png]



	08-09-2006 
	6 h.
	12 h.
	20 h.
	9 h.
	19 h.
	11-09-2006 
	66 h.
	Stripping
	
	
	

	6
	TBC1
	SSS298(3) 
	[image: image8.png]



	09-09-2006 
	56 h.
	24 h.
	28 h.
	6 h.
	0 h.
	14-09-2006 
	114 h.
	Thermal cycle
	
	
	

	7
	TBC1
	SSS298(3) 
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	14-09-2006 
	0 h.
	10 h.
	11 h.
	6 h.
	29 h.
	16-09-2006 
	56 h.
	Stripping
	
	
	

	8
	TBC2
	SSS339(2) 
	[image: image10.png]



	08-09-2006 
	33 h.
	19 h.
	37 h.
	11 h.
	0 h.
	12-09-2006 
	100 h.
	Thermal cycle
	
	
	

	9
	TBC2
	SSS339(2) 
	[image: image11.png]



	12-09-2006 
	0 h.
	18 h.
	41 h.
	8 h.
	14 h.
	15-09-2006 
	81 h.
	Stand By
	
	
	

	10
	TBD1
	SSS526(2) 
	[image: image12.png]



	05-09-2006 
	62 h.
	15 h.
	44 h.
	8 h.
	0 h.
	10-09-2006 
	129 h.
	Thermal cycle
	
	
	

	11
	TBD1
	SSS526(2) 
	[image: image13.png]



	10-09-2006 
	0 h.
	13 h.
	20 h.
	9 h.
	21 h.
	13-09-2006 
	63 h.
	Stripping
	
	
	

	12
	TBE1
	MBAR1374 
	[image: image14.png]



	09-09-2006 
	39 h.
	23 h.
	16 h.
	20 h.
	5 h.
	13-09-2006 
	103 h.
	Stripping
	
	
	

	13
	TBE2
	MBAL1401 
	[image: image15.png]



	10-09-2006 
	13 h.
	46 h.
	16 h.
	14 h.
	9 h.
	14-09-2006 
	98 h.
	Stripping
	
	
	

	14
	TBE2
	MBAR1254 
	[image: image16.png]



	05-09-2006 
	30 h.
	22 h.
	17 h.
	17 h.
	28 h.
	10-09-2006 
	114 h.
	Stripping
	
	
	

	15
	TBF1
	MBAL2168(3) 
	[image: image17.png]



	07-09-2006 
	30 h.
	20 h.
	4 h.
	18 h.
	14 h.
	11-09-2006 
	86 h.
	Stripping
	
	
	

	16
	TBF2
	MBAL1400 
	[image: image18.png]



	05-09-2006 
	40 h.
	30 h.
	25 h.
	13 h.
	21 h.
	11-09-2006 
	129 h.
	Stripping
	
	
	

	17
	TBF2
	MBBL2348 
	[image: image19.png]



	11-09-2006 
	32 h.
	28 h.
	13 h.
	15 h.
	13 h.
	15-09-2006 
	101 h.
	Stripping
	
	
	


 
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10

	1
	MBAR2274
	12669.1 A.
(D1_L)
	12835.7 A.
(D1_L)
	12850 A.
[image: image20.jpg]



	-
	-
	-
	-
	-
	-
	-

	2
	MBAR1338
	12078.8 A.
(D1_U)
	12671.4 A.
(D2_L)
	12848.1 A.
(D2_L)
	[image: image21.png]



	-
	-
	-
	-
	-
	-

	3
	SSS329
	11684 A.
(QE-34)
	12144.2 A.
(QI-12)
	12400 A.
(No Data)
	12394 A.
(QE_34)
	[image: image22.png]



	-
	-
	-
	-
	-

	4
	SSS243
	11598.4 A.
(QE_12)
	12113.2 A.
(QE_12)
	12725 A.
(QI_34)
	[image: image23.png]



	-
	-
	-
	-
	-
	-

	5
	SSS309
	11451.9 A.
(No Data)
	12823 A.
(QE_34)
	12850 A.
[image: image24.jpg]



	-
	-
	-
	-
	-
	-
	-

	6
	SSS298
	10757.9 A.
(QI_12)
	11913.8 A.
(QE_12)
	12163.2 A.
(QE_12)
	12223.2 A.
(QE_34)
	12323.6 A.
(QE_12)
	12155.1 A.
(QE_12)
	12238.1 A.
(QE_12)
	12363.7 A.
(QE_12)
	12510.3 A.
(QE_12)
	[image: image25.png]




	7
	SSS298
	12595 A.
(QE-12)
	12705 A.
(QE-12)
	[image: image26.jpg]



	-
	-
	-
	-
	-
	-
	-

	8
	SSS339
	10642 A.
(No Data)
	10904.9 A.
(QE_34)
	11286.2 A.
(QE-34)
	11526.5 A.
(QI-12)
	11710 A.
(QI-12)
	11827 A.
(QE-34)
	11984 A.
(Qi-34)
	12064.3 A.
(QE_34)
	12167.4 A.
(QE_34)
	12259.4 A.
(QI_34)
	[image: image27.png]




	9
	SSS339
	11163.2 A.
(QI_34)
	11400.2 A.
(QI_34)
	11626.3 A.
(QE_34)
	0 A.
(No Data)
	11797 A.
(QI_34)
	[image: image28.png]



	-
	-
	-
	-
	

	10
	SSS526
	10807.4 A.
(QE_12)
	11243.6 A.
(QE_34)
	12084.1 A.
(QE_34)
	12230.1 A.
(QE_12)
	12437.9 A.
(QE_12)
	12523.8 A.
(QE_34)
	12507.9 A.
(QE_34)
	12608 A.
(QE_34)
	12695.9 A.
(QI_34)
	[image: image29.png]



	

	11
	SSS526
	12213.3 A.
(QE_34)
	11856.5 A.
(No Data)
	12627.4 A.
(QI_12)
	[image: image30.png]



	-
	-
	-
	-
	-
	-
	

	12
	MBAR1374
	11365.4 A.
(D2 - U)
	12224.4 A.
(D1_U)
	12561.7 A.
(D1_U)
	[image: image31.png]



	-
	-
	-
	-
	-
	-
	

	13
	MBAL1401
	11969 A.
(D2-L)
	12416 A.
(D2-U)
	12633.8 A.
(D1- L)
	[image: image32.png]



	-
	-
	-
	-
	-
	-
	

	14
	MBAR1254
	10726.6 A.
(D2_L)
	11710.5 A.
(D1_L)
	12644.3 A.
(D2_L)
	12708.9 A.
(D2_L)
	[image: image33.png]



	-
	-
	-
	-
	-
	

	15
	MBAL2168
	12554.6 A.
(D2-L)
	[image: image34.png]



	-
	-
	-
	-
	-
	-
	-
	-
	

	16
	MBAL1400
	11767.3 A.
(D1_U)
	12081.9 A.
(D2_U)
	12410.3 A.
(D1_U)
	[image: image35.png]



	-
	-
	-
	-
	-
	-
	

	17
	MBBL2348
	12049.9 A.
(D2_L)
	12850 A.
[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
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SSS339 not accepted after 2nd run. 

A lot of thermal cycle.
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_12_M
	TBA2
	MB1338 : NC Generated in CDW2 for resistance value not OK.
	MTM
	
	

	Tuesday_12_M
	TBF1
	MB1374 : Errors in Prep 2.1.
CDW1 OK

Errors in WP04 (NC was generated  for it).

Befor Lyra Tests it was asked to Richard to check WP04.
	MTM
	
	Richard informed that WP04 is OK, and that the file is modified properly. He has uploaded the modified file into MTF and will take action to close the NC.

	Tuesday_12_M
	TBC2
	SSS339: PT6.1 of first Run: First training is a faulty quench?
	MTM
	
	Salem informed :, first training of first run is not a faulty quench. But HF data is not available. So, quench analysis could not be done but Quench location is identified.

	Tuesday_12_M
	TBF1
	MB1331 : D20 cable is cut from the bench. 
	MTM
	
	Richard called. The cable damaged is repaired and checked. 

	Tuesday_12_A
	TBD1
	SSS526: During PT13.1 IAP at warm there was Voltage tap error: VT_Ano_Ext offset=-1000.0mV.
	MTM
	
	Federico contacted. He suspected that some wires may be floating and he told to validate the result and continue. No non conformity required as TC confirmed.

	Tuesday_12_A
	TBF1
	MB1331 :  Comments from equip. Support : 

HV and Res. Meas. are to be repeated after doing IAP because HV test and Res Meas. have to be done only after triggering QH.
	MTM
	
	

	Tuesday_12_A
	TBA1
	MB2274 :.Problem with HF DAQ?
	AB/CO
	HF
	Adriaan Rijllart from AB/CO rectified the problem. He said the HF didn’t get reset and was running in the background. He had to kill the process

	Wednesday_13_M
	TBA1
	MB2274: For the two training quenches Q_loc DAQ was not recorded. 
	MTM
	
	Checks by Richard and Stephane. The connectors of Q_loc in the fixed rack were disconnected.

Card Replaced.

	Wednesday_13_A
	TBB2
	SSS243 : During PT1 IAP permanent triggers. The DAQ could not be armed. 
	MTM
	
	Federico contacted. He found two problems :

Voltage Tap On Internal Bus Bar Open Internally. Bridge put.

Faulty Potaim Card.

Card Replaced.

	Wednesday_13_A
	TBB2
	SSS243: Error in AC compensation test (PT1-part 3).
	MTM
	
	Giorgio attended the problem and suggested to continue.

	Wednesday_13_A
	TBE2
	MB1401 : While launching Training Quench on E2, TEMA stopped & gave message "Configuration files contains ORACLE errors". 
	MTM
	
	Giorgio informed.

	Wednesday_13_A
	TBB2
	During PT4 SPA - File Generated While Power converter switching from On Standby to of while SPA.
	MTM
	
	Giorgio suggested continuing with all tests including Training.



	Wednesday_13_A
	TBB2
	SSS243: TCLAux_X2 error preventing from launching further test. 
	ACR
	
	Cryo team informed.

	Wednesday_13_N
	TBD2
	SSS608 : During training the magnet is quenching when the PC switches from ON STDBY to ON at 44Amps. This has occurred twice due to trigger on MQY1&2_E-I. Unable to analyze the quench. 
	MTM
	
	Giorgio contacted.

	Wednesday_13_N
	TBA1
	MB2274 : Error in PT 13 Offset that could not be compensated.. NC opened.
	MTM
	
	Giorgio contacted The potaim card was changed, and the gain has been corrected in MTF. The NC can be closed.

	Thursday_14_M
	TBB2
	SSS243: Vbus internal trigger during PT8.
	
	
	Giorgio informed. He made some investigation: The two Vtaps of connector P11, A, 1 and 2 are in contact together, but are not in contact with the magnet coil. Giorgio and Richard will perform reflectometry test in order to locate where the circuit is open. NC should be open in CQPT.
Confirmed Thursday afternoon. Giorgio will manage to make possible the last warm up quench.

	Saturday_16_M
	TBB2
	SSS139: No bar code on the cryostat. 
	ICS
	
	NC opened. Giorgio informed. He will take action. Tests can proceed.

	Saturday_16_A

Saturday_16_N
	TBE1 and TBF1
	MB2346, MB1331:DAQ files cannot be copied from local sun to control room

The files generated for clusters TBC1 (SSS338), TBB2 (SSS 139) and TBE2 ( MBAL 2352) need to be uploaded since files are not getting transferred to control room.
	AB/CO
	HF
	Giorgio checked and will inform B. Khomenko. Quench can be analyzed downstairs or the files will be copied to storage/vol12.

Giorgio informed.

	Sunday_17_M
	TBA1
	MB1334: HV test failed on QH vs Ground (Y211). Same QH faulty as in the first run.
	AT/MCS
	
	TC asked to call Gianni Molinari (AT/MCS). G Molinari informed that a responsible from the manufacturer will come on Monday to attend the problem and to keep the magnet in same condition till the time.

	Sunday_17_M
	TBD2
	SSS608: Mismatching in reference values for resistance measurements at warm (actual/old)
	MTM
	
	Giorgio advised to proceed..

































































